% Chris Hanusa
% chanusa@qc.cuny.edu	
\documentclass[12pt]{article} 

\usepackage{amssymb,amsthm,amsmath,epsfig}

\usepackage{color}
\definecolor{darkblue}{rgb}{0, 0, .4}
\definecolor{grey}{rgb}{.7, .7, .7}

\setlength{\topmargin}{-.5in}
\setlength{\textheight}{9in}
\setlength{\textwidth}{6.5in}
%\setlength{\headheight}{26pt}
%\setlength{\headsep}{2pt}
\setlength{\oddsidemargin}{0in}
\setlength{\evensidemargin}{0in}

\newcommand{\lam}{\lambda}

\usepackage[breaklinks]{hyperref} 
\hypersetup{
    colorlinks=true,
    linkcolor=darkblue,
    anchorcolor=darkblue,
    citecolor=darkblue,
    pagecolor=darkblue,
    urlcolor=darkblue,
    pdftitle={},
    pdfauthor={}
}

\begin{document}
\pagestyle{empty}



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{center}\large 
MATH 634, Spring 2014

{\sc Homework 11}

due 5:00{\sc pm} on Monday, March 24.
\end{center}

\noindent
{\em Background reading:} {\em Pearls in Graph Theory}, Section 8.1.

\smallskip\noindent
Problem {\bf 11D} should be typed (or written up) and handed in as class starts on Monday 3/24. You are also expected to be progressing on your project, as described below.

\begin{enumerate}
\item[\bf 11D.]  

\begin{itemize}
\item maximal planar
\item dual graph
\item normal map
\item edge contraction
\item graph minor
\end{itemize}

\end{enumerate}

\bigskip
\noindent{\bf 11W. Project progress.}  For Monday, March 24, I expect everyone to have started working on their Wikipedia project.  In particular, you should have accumulated some sources that you will cite in your work, you should have a plan for working through the project, and you should have started practicing editing pages on Wikipedia in your sandbox.

\medskip
Before class on Monday, email Prof. Chris a link to one (or more) sandbox pages in your Wikipedia user space that includes an outline of your project including a plan of attack for your project.   This will include the following:
\begin{itemize}
\item The scope of your project---what main points will you be focusing on?  (This requires reading texts and deciding on what is important and what is not.)  Break this down using bullet-points to clarify the major topics.
\item You must also include at least two references that have been published in print form, such as books or journals. The copy of the source you consult may be on the internet, but they must have appeared in print form at some point.
\end{itemize}

\noindent
For help getting started using Wikipedia, you should read through the following guide:
\begin{itemize}
\item \url{http://en.wikipedia.org/wiki/Wikipedia:Training/For_students}
\end{itemize}

\noindent
I have started a sandbox that includes everyone's sandbox in our class.
\begin{itemize}
\item \url{https://en.wikipedia.org/wiki/User:GraphTheoryProf/Spring2014}
\end{itemize}

\noindent
As a reminder, here are some helpful places to locate resources:
\begin{itemize}
\item Google Scholar: \url{http://scholar.google.com/}
\item MathSciNet: \url{http://www.ams.org/mathscinet/}
\item The Mathematics Genealogy Project: \url{http://genealogy.math.ndsu.nodak.edu/}
\end{itemize}



\end{document}














